# Modeling Snow Dynamics and Stable Water Isotopes Across Mountain Landscapes

Carroll Rosemary W.H.<sup>1</sup>, Deems Jeffrey S<sup>2</sup>, Sprenger Matthias<sup>3</sup>, Maxwell Reed M.<sup>4</sup>, Brown Wendy S<sup>5</sup>, Newman Alexander<sup>5</sup>, Beutler Curtis A<sup>5</sup>, and Williams Kenneth Hurst<sup>6</sup>

<sup>1</sup>Desert Research Institute <sup>2</sup>University of Colorado Boulder <sup>3</sup>Lawrence Berkeley National Laboratory <sup>4</sup>Princeton University <sup>5</sup>Rocky Mountain Biological Laboratory <sup>6</sup>Lawrence Berkeley National Laboratory (DOE)

November 16, 2022

#### Abstract

A coupled hydrologic and snowpack stable water isotope model assesses controls on isotopic inputs across a large, mountainous basin. The most depleted isotope conditions occur in the upper subalpine where snow accumulation is high and rainfall is low. Snowmelt evolution over time indicates isotopic enrichment is not dictated by melt fractionation but is determined by elevation which controls the amount, phase and isotopic mass of spring precipitation coincident with the ablation period. With respect to snowpack kinetic fractionation, its effect on snowmelt is a balance between energy and snow-availability. It is highest above treeline and in the shrub-dominated upper montane where vegetation shading is low, while deep snowpack and conifer forests limit the influence of kinetic fractionation in the subalpine. Wet years reduce the effects of snowpack fraction on snowmelt across the basin, except in the lower montane where added snowfall bolsters water-limited conditions.

#### Hosted file

essoar.10510911.1.docx available at https://authorea.com/users/363284/articles/595436-modeling-snow-dynamics-and-stable-water-isotopes-across-mountain-landscapes

### Hosted file

carroll\_2022\_grl\_si.docx available at https://authorea.com/users/363284/articles/595436modeling-snow-dynamics-and-stable-water-isotopes-across-mountain-landscapes Rosemary W.H. Carroll<sup>1</sup>, Jeffrey Deems<sup>2</sup>, Matthias Sprenger<sup>3</sup>, Reed Maxwell<sup>4</sup>, Wendy Brown<sup>5</sup>, Alexander Newman<sup>5</sup>, Curtis Beutler<sup>5</sup>, Kenneth H. Williams<sup>3,5</sup>

<sup>1</sup>Desert Research Institute, Reno, NV; <sup>2</sup>National Snow and Ice Data Center, Boulder, CO; <sup>3</sup>Lawrence Berkeley National Laboratory, Berkeley, CA, <sup>4</sup>Princeton University, Princeton, NJ; <sup>5</sup>Rocky Mountain Biological Laboratory, Gothic CO

Corresponding author: Rosemary Carroll (rosemary.carroll@dri.edu)

Key Points:

- The most depleted isotopic water inputs occur in the upper subalpine where snow accumulation is high and rainfall is low.
- Deep snowpack and shading of conifer forests limit the influence of kinetic fractionation (sublimation) of snowpack in the subalpine.
- Effects of kinetic fractionation decreases in big snow years except in the lower montane where added snowfall reduces water-limitation.

## Abstract

A coupled hydrologic and snowpack stable water isotope model assesses controls on isotopic inputs across a large, mountainous basin. The most depleted isotope conditions occur in the upper subalpine where snow accumulation is high and rainfall is low. Snowmelt evolution over time indicates isotopic enrichment is not dictated by melt fractionation but is determined by elevation which controls the amount, phase and isotopic mass of spring precipitation coincident with the ablation period. With respect to snowpack kinetic fractionation, its effect on snowmelt is a balance between energy and snow-availability. It is highest above treeline and in the shrub-dominated upper montane where vegetation shading is low, while deep snowpack and conifer forests limit the influence of kinetic fractionation in the subalpine. Wet years reduce the effects of snowpack fraction on snowmelt across the basin, except in the lower montane where added snowfall bolsters water-limited conditions.

#### Plain Language Summary

Stable water isotopes have been use for decades in hydrology to track vegetation water use, groundwater recharge and stream water source. Watersheds reliant on snow alter the timing of water inputs through snow storage and melt and may produce a different isotopic input signal due to evaporation of the snowpack prior to melt. We combine a hydrologic and snowpack isotope model to understand how landscape position and climate may affect isotopic water inputs in a large mountain basin with nearly 2 km in vertical relief. The heaviest isotopes occur in the upper subalpine where snow accumulation is highest and rain inputs are low. The temporal evolution of isotopes in snowmelt is controlled by elevation and its influence on the amount, phase (rain or snow) and isotopic mass of spring precipitation coincident with the snowmelt period. Changes to the snowpack isotopic signature by vapor loss are most important where vegetation does not shade the snow, where moderate snowfall occurs and evaporation potential is relatively high. Changes are highest above treeline and in the shrub-dominated upper montane. Less change occurs in the deep snow found in the conifer forests of the subalpine, and in the snow-limited lower montane.

#### 1 Introduction

Mountain snowpack is an important water resource globally (Barnett et al., 2005) and is especially vulnerable to climate change (Hock et al., 2019; Huning & AghaKouchak, 2020; Milly et al., 2008). Despite the importance of snowdominated systems, uncertainty on how snow becomes streamflow remains (Julander & Clayton, 2018). Stable isotopes of water  $({}^{18}O/{}^{16}O, {}^{2}H/{}^{1}H)$  have long been used as tracers in hydrologic research to identify the partitioning of rain and snow to vegetation water use (Berkelhammer et al., 2020; Sprenger et al., 2016), groundwater recharge (Earman et al., 2006; Fiorella et al., 2018; Jasechko, 2019: Oiro et al., 2018), and stream water (Cowie et al., 2017: Zhang et al., 2018). Isotopic mass balance studies in watersheds reliant on snow water must adjust isotopic boundary influxes as a function of snow storage and snowmelt timing, and the potential for isotopic modification due to post-depotional fractionation in the snowpack. While isotopic fractionation processes in snow are fairly well defined (section 1.1), less work has addressed these processes at watershed-scales in mountain environments. Challenges are largely due to obtaining hydrologic and isotopic observations at the scales important to snow processes (Bales et al., 2006; Clark et al., 2011; Mott et al., 2018). Temporal scales are defined by meteoroogical inputs (hourly-daily) needed to quantify the energy balance of the snowpack, while the spatial resolution needed to capture non-uniform hydrologic processes in mountain systems is on the order of 100 to 250 m (Baba et al., 2019; Foster et al., 2020). An added complication arises given most of snow resides near treeline (Carroll et al., 2019) with regular and safe access for field samples often not possible. As a consequence, isotopic tracer studies in mountain environments tend to extrapolate limited plot-scale, snowpit data across an entire watershed (Bearup et al., 2014; Carroll et al., 2018; Fang et al., 2019; Evans et al., 2016) with the potential to introduce significant error in stream water source estimates dependent on isotopic inputs.

With respect to time-variable isotopic inputs from snowmelt, these have largely been confined to laboratory experiments (Feng et al., 2002; Taylor et al., 2002) or to field studies focused on snowpit (Friedman et al., 1991; Stichler et al., 1981; Taylor et al., 2001) or hillslope scales (Evans et al., 2016). More recently, Ala-aho et al. (2017) incorporated changes in snow isotopic values with a snow process model to estimate a spatially distributed snowmelt isotope signal into basins with contrasting snow conditions. We apply work by Ala-aho et al. (2017) across a much larger Colorado River headwater basin with greater relief. We constrain this model using a comprehensive isotopic dataset in precipitation, snowpack and snowmelt (Carroll et al., 2022). Through this data-modeling framework, we investigate water isotopic inputs at high spatial and temporal resolution over multiple years with reference to snow processes. Our objective is to isolate key controls on isotopic water inputs and better constrain where and when post-depositional fractionation is important. Results will inform data needs for hydrologic tracer-based studies within mountain environments.

#### 1.1 Stable Water Isotope Overview

Stable water isotopes are reported as the ratio of heavier to lighter isotopes in a sample  $(R_{sample} = {}^{18}O/{}^{16}O, \text{ or } {}^{2}H/{}^{1}H)$  relative to the Vienna Standard Mean Ocean Water standard  $(R_{VSMOW})$ . Results are presented in units of per mil (‰) as  $\delta = 1000 * (R_{\text{sample}} - R_{\text{VSMOW}})/R_{\text{VSMOW}}$ . The second order isotope parameter, d-excess (d-excess =  ${}^{2}\text{H} + 8 {}^{18}\text{O}$ ) expresses the deviation of local samples from the Global Meteroic Water Line (GMWL) as plotted in the dualisotope space defined by  ${}^{2}\text{H}$  versus  ${}^{18}\text{O}$ . D-excess decrease is in response to kinetic processes resulting from the molecular mass differential between oxygen and hydrogen during vapor loss. Specifically, the lighter  ${}^{2}\text{H}$  molecule turns to vapor more readily than the heavier  ${}^{18}\text{O}$  molecule. A lowering of d-excess from its initial isotopic state indicates either evaporation or sublimation (Clark & Fritz, 1997).

In this study, we define the initial state as the isotopic composition of precipitation. Precipitation inputs depend on temperature, relative humidity, origin of the air mass (Clark & Fritz, 1997). For the continental interior of the United States, snowfall originates from northwest frontal storms (Marchetti & Marchetti, 2019) and contains depleted isotopic values due to cold, high elevation conditions with a low vapor fraction (Dansgaard, 1964). In contrast to snow, rain tends to experience more recycling of moisture via evaporation than winter precipitation as storms move inland from the ocean. Consequently, rainfall contains more enriched isotopic conditions and is comparatively lower in d-excess.Precipitation isotopic inputs are also affected by elevation with <sup>18</sup>O lapse rates in North America ranging from -0.17 to -0.22% per 100 m in elevation (Friedman et al., 1992; Tappa et al., 2016). After deposition, the ratios of heavy to light isotopes in the snowpack can vary due to diffusional transport of water from the soil, temperature-gradient induced vapor diffusion within the snow column, lateral flow through the snowpack and fractionation processes associated with sublimation, evaporation, and melt-freeze cycles (Beria et al., 2018; Cooper, 1998; Evans et al., 2016; Friedman et al., 1991; Sinclair & Marshall, 2008; Stichler et al., 1981).

#### 2 Site Description and Modeling Strategy

The East River, Colorado (ER, 750 km<sup>2</sup>, Figure 1a) is a headwater basin of the Colorado River in the southwestern United States with elevations ranging from 2440 to 4346 m. Climate for the area is continental subarctic with long, cold winters and short, cool summers. Annual precipitation  $1,434 \pm 258 \text{ mm/yr}$ (years 1987-2020), with  $25\pm7\%$  falling as rain. Most rain occurs during the summer monsoon (Carroll et al., 2020). Water year 2015 had the most annual rainfall to toal precipitation (40%) and 2017 had the least (12%). June is typically very dry with little precipitation. Water years considered in this study (2015-2020) are coincident with isotopic surveys in the ER (Carroll et al., 2022). Water year 2016 represents average snow conditions, and 2018 and 2019 represent dry and wet snow conditions, respectively. For the years studied, winter temperatures were highest in 2017 and 2018, while 2019 represents the coolest winter. Ecozones are broadly defined by elevation and dominant vegetation cover (Figure 1b). Montane conditions (<2800 m) are dominated by shrubs, grasses and forbs. The subalpine is divided into the lower subalpine with dense conifer forests mixed with aspen (2800-3200 m), and the upper subalpine which is a combination of of low density conifer forests, shrubs and barren ground (3200-3500 m). The alpine (>3500 m) occurs above treeline.

The hydrologic model uses the semi-empirical, spatially distributed U.S. Geological Survey (USGS) numerical code Precipitation-Modelling Runoff System (PRMS, Markstrom et al., 2015). Water and energy are tracked daily through the atmosphere, canopy and subsurface at a 100 m grid resolution. Daily climate forcing assigns minimum and maximum temperature lapse rates based on two local snow-telemetry stations and adjusted for aspect. Potential evapotranspiration (PET) is calculated with a modified version of the Jensen-Haise formulation dependent on air temperature and solar radiation (Jensen et al., 1969). Solar radiation is based on a modified degree-day method developed in the Rocky Mountain region and applicable for sites with clear skies on days that lack precipitation (Leavesley et al., 1983). Following methods presented in previous work (Carroll et al., 2019), the spatial distribution of snow water equivlant (SWE), based on airborn light-detection and ranging (LiDAR) (Painter et al., 2016). The approach implicitly allows redistribution by wind and avalanche to move snow off mountain ridges and into high elevation circue valleys to produce the deepest snowpack near treeline. Rain was distributed using the Parameterelevation Relationships on Independent Slopes Model (PRISM, 800 m) 30-year monthly averages for 1981-2010 (OSU, 2012). Figure 1c illustrates predicted SWE near peak accumulation in a wet water year. Vegetation cover at the 1 m resolution (Breckheimer, 2021) was overlain with the USGS Landfire (2015) 30 m resolution meadow and then resampled to the 100 m grid. PRMS parameters for summer and winter cover density, canopy interception characteristics for snow and rain, and transmission coefficients for short wave solar radiation follow (Gardner et al., 2018).

Precipitation phase (rain, snow) for a given location is controlled by a userdefined temperature threshold. Precipitation contributes water to the snowpack and adds/subtracts energy content based on its phase and temperature. Shortwave radiation at the snowpack surface is limited by the winter vegetation transmission coefficient and reduced by the estimated albedo. Trees and shrubs are assumed to diminish wind, and the energy applied to the snowpack in these areas is diminished by half. Sublimation is calculated with a user-defined fraction of PET adjusted for heat deficit in the snowpack and snow-covered area. Refer to Carroll *et al.*, (2022) for model calibration.

Snowpack isotopic samples were collected years 2016 to 2020 to span gradients in

topography, vegetation and seasonal climate. Overlapping isotopic campaignes for precipitation and snowmelt were conducted across elevation. Refer to Carroll et al. (2022) for detailed analysis of data. Observed isotopic values in snowfall are defined by air temperature and occur along the GMWL. More enriched snowfall occurs when temperatures are warmer meaning snowfall in the spring has higher <sup>18</sup>O than in the middle of winter. Observed isotopic values in rainfall are also dependent on air temperature but are modified by wind speed and barometric pressure. Rain is more enriched than snowfall for an equal temperature and falls below the GMWL (d-excess is lower). The observed isotopic elevational lapse rate is -0.16% <sup>18</sup>O per 100-m. The isotope mass balance model is a parsimonious approach that follows Ala-aho et al. (2017) to track <sup>18</sup>O entering the soil system as snowmelt or rain. The approach was expanded to include <sup>2</sup>H and and kinetic fractionation. Water stores and fluxes needed for the isotope model use hydrologic model output for each timestep and model grid location. Similar to Ala-aho et al. (2017), isotopic parameters related to fractionation are estimated using a Monte Carlo approach with uniform input distirbutions and 1000 realizations. Modeled fit is based on a relative root mean squared error (rRMSE) for each isotope parameter (e.g. <sup>2</sup>H, <sup>18</sup>O, d-excesss) and type of observation (e.g. snowpack, snowmelt). The final parameter suite for melt, evaporative and kinetic fractionation is determined from the average of the 10-best realizations with the lowest composite rRMSE. A full description of the snowpack isotopic model and its calibration is provided in the Supporting Information (SI).

# 4 Results

The spatial distribution of annual <sup>18</sup>O influxes from snowmelt and rain for average water year conditions is given in Figure 1d. Excluding 2015, the most depleted values (-17.8±0.4‰) occur in the transition between the upper subalpine and alpine (3500 m) where the snow fraction (0.90±0.04) and spring snow water inputs are largest. Relative enrichment in heavy isotopes occurs in the alpine (-15.6±1.2‰) and montane (-12.5±1.1‰). Some enrichment also occurs along the west-to-east trajectory largely related to estimated summer rainfall. Water year 2015 was extremely warm, received much of its snow in May and experienced twice the annual average of its moisture as rain. This shifts <sup>18</sup>O to more enriched conditions (+3.8±0.5‰). Canopy interception losses are tracked and account for significant water loss back to the atmosphere (annually=4±3%; Oct-May=8±3%). The modeled effect is to bias toward more depleted snowmelt with the effect greatest in the lower subalpine and montane. Annually the bias of canopy loss on isotopic water influx to the soil is -1.6‰ and for winter only is -0.8‰.

Melt fractionation is estimated a small effect in the ER. However, snowmelt evolution over time indicates the enrichment rate is largely associated with the amount and timing of spring precipitation coincident with the amount of SWE over the ablation period. To illustrate, daily SWE, snowmelt and snowmelt <sup>18</sup>O are provided for two locations (Figure 2). The higher elevation site is in

a conifer forest of the upper subalpine with a north aspect (3361 m). This site experiences deep and persistent snowpack with snowmelt delayed until mid-April and lasting into early July. The highest melt rates occur during the dry month of June. Initial  ${}^{18}$ O in snowmelt is -20.4% and enriches by 2.5% at snowpack disappearance on 1 July. Prior to 26 June snowmelt <sup>18</sup>O exhibits a non-linear, direct relationship to snowmelt ( $r^2=0.58$ , p(0.01)). Initial snowmelt enriches quickly as a function of snowmelt rate but plateaus at -19.3% when snowmelt exceeds 5 mm/d. On average, the enrichment rate is 0.017% per day. Snowfall occurs on 25 June adding enriched late spring snow to a diminished snowpack (SWE=40 mm). This forces a jump in the remaining snowmelt by +1%. With no added snowfall, snowpack and snowmelt rates decline with snowmelt enriching 0.06% per day with this very enriched snowmelt only 6%of the snowmelt mass from initial melt onset. The second, and lower elevation, site is located in a montane meadow (2620 m). Maximum SWE is half the subalpine example. Snowmelt begins on 18 February (-20.6%) and enriches 3.9% when seasonal snowpack is gone on 25 April. SWE declines despite significant snowfall in March and April, with increased snowmelt rate describing 42% of the enrichment rate (p«0.01) prior to 17 April, and snowmelt enrichment 0.038% per day. Significant late-April snowfall occurs under warm conditions (snow <sup>18</sup>O inputs  $-14.1 \pm 6.0\%$ ). Enriched snowfall contributes significantly to the snowpack when SWE is low, generates a substantial jump in snowmelt <sup>18</sup>O (+2%) and maintains the same direct relationship to snowmelt rate and <sup>18</sup>O influx. Once SWE=0, ephermerial snow <sup>18</sup>O contributions are variable based on seasonal temperature with rain <sup>18</sup>O much more enriched than snow.

Kinetic fractionation in snowpack is tracked by decreases in d-excess compared to incoming precipititation. The largest decreases in d-excess in comparison to precipitatin inputs occur swhere there is a moderately high amount of available energy (PET) compared to precipitation (P) (PET/P =1.5, Figure 3d). This occurs in upper portions of the montane. At higher elevations, where energy is limited (PET/P<1), changes in d-excess are dictated by vegetation type ( $r^2=0.62$ ,  $p\ll0.01$ ) with the ranking of highest-to-lowest d-excess declines occurring at locations containing meadow, bare rock, aspen, shrub and conifer. An indirect relationship to snow fraction acts as a secondary control but its predictive power is weak ( $r^2=0.08$ ,  $p\ll0.01$ ). For locations in the ER where water is more limited (PET/P>1.5), lower snowfall fraction reduces the influence of snowpack kinetic fractionation ( $r^2=0.15$ ,  $p\ll0.01$ ). Cover type ( $r^2=0.06$ ) and increased solar radiaton ( $r^2=0.06$ ) add only modest descriptive ability.

Interannually, the relative amount of kinetic fractionation in snowpack increases when total precipitation is low but snow fraction is high and solar radiation is high ( $r^2=0.89$ ). A comparison of d-excess declines is provided in Figure 3 for a dry/warm water year (2018) and a wet/cool water year (2019). The influence of kinetic fractionation in snowpack decreases across most of the basin in 2019 with the largest reductions in kinetic fractionation occurring in the shrub-dominated upper montane. Very little change occurs in the subalpine. Increases in kinetic fractionation occur only in the lower montane. Hydrologically, water year 2019 snowfall is 183% greater than 2018, with snow covered area larger and more persistent, lasting two additional months. Daily sublimation losses are similar for both winters, based on energy limited conditions, until late April when sublimation losses in 2019 exceed those in 2018 as a function of snow persistence. By years-end sublimation loss in 2019 is 23% larger than 2018 but the basin average sublimation to snowfall in 2019 is half that computed for 2018.

#### 5 Discussion

The spatial variability in simulated annual <sup>18</sup>O water inputs from snowmelt and rain is dominated by the isotopic composition of precipitation (Dahlke & Lyon, 2013; Stichler et al., 1981) which is based on air temperature, altitude and precipitation phase (Carroll et al., 2022; Clark & Fritz, 1997; Dansgaard, 1964; Otte et al., 2017). Cold season snowfall is isotopically depleted in comparison to rainfall to produce large seasonal isotopic swings. Therefore the timing of precipitation matters, and water years like 2015, that received a significant amount of snow in May and an exceptional amount of rainfall is estimated to produce a much lighter annual isotopic input signal to the soil compared to other years simulated. Elevation is another critical control producing a 2.9%variation in <sup>18</sup>O across the ER based on the observed lapse rate. Elevation also determines precipitation phase and accolocated snow fraction of annual water inputs. The largest volumes of snowmelt occur in the upper subalpine, where deep and persistent snowpack accumulates over the cold winter months to produce a large pulse of the most depleted water entering the basin. At higher elevations in the alpine environment, snow redistribution moves snow off ridges toward treeline. The diminished snowpack is combined with large quantities of enriched summer rain from orographically-driven monsoon events to reduce the annual snow fraction and promote more enriched isotopic inputs compared to the conifer regions. Isotopically light water inputs to the montane from more enriched snowfall is further enriched by a much lower snow fraction due to temperature-driven phase shifts producing more rain in the fall and spring. Quantifying the isotopic lapse rate, the amount of precipitation and where and when precipitation is rain or snow are critically important to estimating the annual mass flux of <sup>18</sup>O into the terrestrial system.

Studies focused on intercepted snow by canopy in the intermountain western United States indicate <sup>18</sup>O enrichment of stored snow is approximately 2.1‰ with rates of enrichment increasing for smaller snow particles, denser canopy, longer residence times and under clear-sky conditions (Claassen & Downey, 1995; Koeniger et al., 2008). Exchange of enriched snow from the canopy to the ground could have an influence on water inputs to the soil. For example, Ala-aho *et al.* (2017) simulated 0.7‰ enrichment in <sup>18</sup>O in ground snowpack with a three-fold increase in leaf-area index. We find canopy interception loss is consequential on the annual water balance of the ER (16±4%) and comparable to other estimates in north-central Colorado (Sexstone et al., 2018). We do not simulate the exchange of eriched snow stored in the canopy to the underlying snowpack, but we do account for the isotopic mass balance of underlying snowpack based

on canopy evaporation. When atmospheric demand is high, canopy storage is continually freed-up for more interception. This occurs in the lower subalpine and montane where either denser conifer forests or high PET occur, and rate of possible vapor loss is higher in the summer than winter. The emphasis on summer canopy capture of rain is to bias annual ground influxes of <sup>18</sup>O toward more depleted winter values. This is not necessarily incorrect, but future work will need to assess if enrichment of snowpack by throughfall is important or if forest shading (discussed below) is a more important control on snowpack isotopic content.

The temporal evolution of simulated <sup>18</sup>O in snowmelt for the ER follows well established trends that reflect mass conservation in the snowpack during ablation with initial melt water isotopically depleted in <sup>18</sup>O in comparison to snowpack and enriching over time (Ala-aho et al., 2017; Beria et al., 2018; Taylor et al., 2002; Taylor et al., 2001). Total enrichment in snowmelt is similar to observed trends in other studies (Lee et al., 2010; Unnikrishna et al., 2002) with total snowmelt enrichment and faster rates of enrichment occurring at lower elevations (Carroll et al., 2022). However, model results suggest ER snowmelt enrichment rates are not dictated by melt fractionation. Ala-aho et al. (2017) found a similar result for Bogus Creek in Idaho and suggested this may reflect preferential meltwater flow and limited interaction with the bulk snowpack (Evans et al., 2016; Unnikrishna et al., 2002).

We propose, the timing of snow accumulation and melt across mountain gradients is more consequential on snowmelt isotopic evolution than melt fractionation. From empirical evidence, Carroll et al., (2022) suggests initially depleted snowmelt at low elevations is related to delayed snowpack accumulation when seasonal precipitation is more depleted. Model results suggest this depleted initial melt signature is also a consequence of earlier melt onset when the snowpack is still fairly depleted. Melt enrichment is faster at lower elevation as a consequence of the precipitation lapse rate contributing a larger mass of <sup>18</sup>O over the spring compared to higher elevatations and doing so when melt has begun. The result is a direct relationship between snowmelt enrichment and melt rate. At high elevations, the largest melt rates occur in June when there is no added spring snow or rain. As a result, snowmelt enrichment tracks bulk <sup>18</sup>O mass accured over the entire snow season and appears rate-limited with respect to melt. At both locations, very shallow snowpack (<100 mm) is influenced by spring/summer snowfall to dramatically enrich snowmelt. Voumetrically, the effect is more significant to the overall water influx from seasonal snowpack at lower elevations.

D-excess reductions in snowpack are explored as functions of landscape position and climate to provide guidance to where and when isotopic sampling of snowpack needs to consider kinetic fractionation in its calculation of isotopic inputs to the soil. D-excess delines are also an indicator of where sublimation is a critical component of the water balance. Sublimation is affected by energy and snow availability and is highest were both are maximized (PET/P~1). Model results find d-excess reductions in snowpack described by decreased precipitation, increased snow fraction and increased solar radiation. The combination of all three variables minimizes the snowpack such that sublimation is a larger proporation of the snow budget, but maximizes snow persistence and energy. The winters in the ER are energy-limited and while increased sublimation may occur in a big snow year due to increased snow persistence, it is not globally sufficient to increase the effect of post-depositional kinetic fraction over large increases in SWE. Model results find the spatial distribution of post-depositional fractionation is highest where snow fraction is moderate and vegetation shading is low, with secondary controls at lower elevations related to PET and solar radiation. Consequently the largest effect of snowpack fractionation occurs in the upper montane, and to a slightly lower degree in the alpine.

Effects of snowpack fractionation are estimated very low in the deep snowpack and conifer forests of the subalpine across all climate conditions. Conifer forests greatly reduce wind scour (Elder et al., 1991) to promote deeper snowpack to buffer d-excess delines via mass balance. Conifer forests also reduce solar radiation on the snowpack surface (Musselman, Molotch and Brooks, 2008; Molotch et al., 2009; Varhola et al., 2010) to limit evaporative losses. Snow water-limitation in the lower montane also reduces the potential for fractionation, except on wet and cool years when snow persistence is sufficient that sublimation increases and d-excess declines are more significant. Globally, postdepositional metamorphism is in the energy-limited ER is not a not a major consideration given the bulk of snow water entering the system resides in the subalpine, but detailed hydrologic analysis above treeline or in the montane needs to consider snowpack fractionation in isotopic tracer-based analyses to reduce error.

#### 6 Conclusions

We combined a hydrologic and snowpack isotopic fractionation model with a previously published comprehensive isotopic data set to assess stable water isotope influxes into a large mountainous watershed of large relief. Modeled inputs are assessed at the temporal and spatial scales pertinent to quantifying energy and water balance of snowpack in complex terrain. Influxes of  $^{18}$ O from snowmelt and rain are dominated by the seasonal variability in precipitation amount, phase and isotopic value with the most depleted inputs occurring in the upper subalpine where snow accumulation is highest and rain inputs minimal. Total enrichment and rate of enrichment in <sup>18</sup>O is higher at lower elevations based on higher mass of <sup>18</sup>O in precipitation, earlier onset of melt and ongoing spring snowfall coincident with the melt of seasonal snowpack. Post-depositional kinetic fractionation by vapor loss is most important where and when persistent but low snowpack can sustain sufficient sublimation. It is most important in the barren alpine environment (above treeline) and the shrub-dominated upper montane. It is least important in the subalpine where snowpack is deep and is shaded by conifer forests. Given the ER is largely energy-limited, wet water years reduce the effect of snowpack kinetic fractionation across the basin. The exception is in the lower montane where snow-limited conditions are moderated by the added snowfall.

## Acknowledgments and Data

The snowpack isotope model is provided in the Supporting Information. Refer to Carroll et al. (2020) for details on the hydologic model. Isotope data files are available to the public (https://data.ess-dive.lbl.gov/view/doi: 10.15485/1824223). Funding from the US Department of Energy, Office of Science under contract DE-AC02-05CH11231 through Lawrence Berkeley National Laboratory.

#### **References Cited**

Ala-aho, P., Tetzlaff, D., McNamara, J. P., Laudon, H., Kormos, P., & Soulsby, C. (2017). Modeling the isotpic evolution of snowpack and snowmelt: testing a spatially distributed parsimonious approach. Water Resources Research, 53(5813-5830). https://doi.org/DOI:10.1002/2017WR020650Baba, M. W., Gascoin, S., Kinnard, C., Marchane, A., & Hanich, L. (2019). Effect of Digital Elevation Model Resolution on the Simulation of the Snow Cover Evolution in the High Atlas. Water Resources Research, 55(7), 5360-5378. https://doi.org/10.1029/2018WR023789Bales, R. C., Molotch, N. P., Painter, T. H., Dettinger, M. D., Rice, R., & Dozier, J. (2006). Mountain hydrology of the western United States. Water Resources Research, 42(8), 1-13. https://doi.org/10.1029/2005WR004387Barnett, T. P., Adam, J. C., & Lettenmaier, D. P. (2005). Potential impacts of a warming climate on water availability in snow-dominated regions. Nature, 438, 303–309. https://doi.org/http://dx.doi.org/10.1038/nature04141.Bearup, L. a, Maxwell, R. M., Clow, D. W., & Mccray, J. E. (2014). Hydrological effects of forest transpiration loss in bark beetle-impacted watersheds. Nature Climate Change, 4(6), 481-486. https://doi.org/10.1038/NCLIMATE2198Beria, H., Larsen, J. R., Ceperley, N. C., Michelon, A., Vennemann, T., & Schaefli, B. (2018). Understanding snow hydrological processes through the lens of stable water isotopes. Wiley Interdisciplinary Reviews: Water. https://doi.org/10.1002/wat2.1311Berkelhammer, M., Still, 5(6). 1-23. C., Ritter, F., Winnick, M., Anderson, L., Carroll, R., Carbone, M., & Williams, K. H. (2020). Persistence and plasticity in conifer water-Journal of Geophysical Research: Biogeosciences, 125(2). use strategies. https://doi.org/https://doi.org/10.1029/2018JG004845Breckheimer, I. (2021). High resolution landcover maps for the upper Gunnison Basin derived from LiDAR and NAIP imagery. Watershed Function Science Focus Area. https://doi.org/doi:xx.xxxx/xxxxxCarroll, R.W.H., Deems, J., Maxwell, R., Sprenger, M., Brown, W., Newman, A., Beutler, C., Bill, M., Hubbard, S. S., & Williams, K. H. (2022). Variability in observed stable water isotopes in snowpack across a mountainous watershed in Colorado. Hydrological Processes, in review.Carroll, Rosemary W. H., Deems, J. S., Niswonger, R., Schumer, R., & Williams, K. H. (2019). The Importance of Interflow to Groundwater Recharge in a Snowmelt-Dominated

Headwater Basin. Geophysical Research Letters, 46(11), 5899–5908. https://doi.org/10.1029/2019GL082447Carroll, Rosemary W.H., Bearup, L. A., Brown, W., Dong, W., Bill, M., & Williams, K. H. (2018). Factors controlling seasonal groundwater and solute flux from snow-dominated basins. Hydrological Processes, 32(14), 2187–2202. https://doi.org/10.1002/hyp.13151Carroll, Rosemary W.H., Deems, J. S., Niswonger, R., Schumer, R., & Williams, K. H. (2019). The Importance of Interflow to Groundwater Recharge in a Snowmelt-Dominated Headwater Basin. Geophysical Research Letters, https://doi.org/10.1029/2019GL082447Carroll, Rose-46(11), 5899-5908.mary W.H., Gochis, D., & Williams, K. H. (2020). Efficiency of the Summer Monsoon in Generating Streamflow Within a Snow-Dominated Headwater Basin of the Colorado River. Geophysical Research Letters, 47(23). https://doi.org/10.1029/2020GL090856Claassen, H. C., & Downey, J. S. (1995). A model for deuterium and Oxygen-18 isotope changes during evergreen interception of snowfall. Water Resources Research. 31(3). https://doi.org/https://doi.org/10.1029/94WR01995Clark, I. D., 601 - 618.& Fritz, P. (1997). Environmental Isotopes in Hydrogeology. Lewis Publishing.Clark, M. P., Hendrikx, J., Slater, A. G., Kavetski, D., Anderson, B., Cullen, N. J., Kerr, T., Örn Hreinsson, E., & Woods, R. A. (2011). Representing spatial variability of snow water equivalent in hydrologic and land-surface models: A review. Water Resources Research, 47(7). https://doi.org/10.1029/2011WR010745Clark, Martyn P., Hendrikx, J., Slater, A. G., Kavetski, D., Anderson, B., Cullen, N. J., Kerr, T., Örn Hreinsson, E., & Woods, R. A. (2011). Representing spatial variability of snow water equivalent in hydrologic and land-surface models: A review. Water Resources Research, 47(7). https://doi.org/10.1029/2011WR010745Cooper, Isotopic Fractionation in Snow Cover. L. W. (1998). Isotope Tracers in Catchment Hydrology, 119–136. https://doi.org/10.1016/b978-0-444-81546-0.50011-2Cowie, R. M., Knowles, J. F., Dailey, K. R., Williams, M. W., Mills, T. J., & Molotch, N. P. (2017). Sources of streamflow along a headwater catchment elevational gradient. Journal of Hydrology, 549(April), 163–178. https://doi.org/10.1016/j.jhydrol.2017.03.044Dahlke, H. E., & Lyon, S. W. (2013). Early melt season snowpack isotopic evolution in the Tarfala valley, northern Sweden. Annals of Glaciology, 54(62), 149-156.https://doi.org/10.3189/2013AoG62A232Dansgaard, W. (1964). Stable isotopes in precipitation. Tellus, 16(4), 436-468. https://doi.org/10.3402/tellusa.v16i4.8993Earman, S., Campbell, A. R., Phillips, F. M., & Newman, B. D. (2006). Isotopic exchange between snow and atmospheric water vapor: Estimation of the snowmelt component of groundwater recharge in the southwestern United States. Journal of Geophysical Research Atmospheres, 111(9), 1-18. https://doi.org/10.1029/2005JD006470Elder, K., Dozier, J., & Michaelsen, J. (1991). Snow accumulation and distribution in an alpine watershed. Water Resources Research, 27, 1451–1552. Evans, S. L., Flores, A. N., Heilig, A., Kohn, M. J., Marshall, H. P., & McNamara, J. P. (2016). Isotopic evidence for lateral flow and diffusive transport, but not sublimation, in a sloped seasonal snowpack, Idaho, USA. Geophysical Research

Letters, 43(7), 3298-3306. https://doi.org/10.1002/2015GL067605Fang, Z., Carroll, R. W. H., Schumer, R., Harman, C., Wilusz, D., & Williams, K. H. (2019). Streamflow partitioning and transit time distribution in snowdominated basins as a function of climate. Journal of Hydrology, 570(December 2018), 726–738. https://doi.org/10.1016/j.jhydrol.2019.01.029Feng, X., Taylor, S., Renshaw, C. E., & Kirchner, J. W. (2002). Isotopic evolution of snowmelt 1. A physically based one-dimensional model. Water Resources Research, https://doi.org/10.1029/2001WR000814Fiorella, R. P., 38(10), 35-1-35-38.Poulsen, C. J., & Matheny, A. M. (2018). Seasonal Patterns of Water Cycling in a Deep, Continental Mountain Valley Inferred From Stable Water Vapor Isotopes. Journal of Geophysical Research: Atmospheres, 123(14), https://doi.org/10.1029/2017JD028093Foster, L. M., Williams, 7271 - 7291.K. H., & Maxwell, R. M. (2020). Resolution matters when modeling climate change in headwaters of the Colorado River. Environmental Research Letters, 15(10).https://doi.org/10.1088/1748-9326/aba77fFriedman, I., Smith, G. I., Gleason, J. D., Warden, A., & Harriss, J. M. (1992). Stable isotope composition of waters in Southeastern California 1. Modern precipitation. Journal of Geophysical Research, 97, 5795–5812. Friedman, Irving, Benson, C., Gleason, J. I. M., & Survey, S. G. (1991). Isotopic changes during snow metamorphism oD oD. 3.Gardner, M. A., Morton, C. G., Huntington, J. L., Niswonger, R. G., & Henson, W. R. (2018). Input data processing tools for the integrated hydrologic model GSFLOW. Environmental Modelling and Software, 109(June), 41-53. https://doi.org/10.1016/j.envsoft.2018.07.020Hock, R., Rasul, G., Adler, C., Cáceres, B., Gruber, S., Hirabayashi, Y., Jackson, M., Kääb, A., Kang, S., Kutuzov, S., Milner, A., Molau, U., Morin, S., Orlove, B., & Steltzer, H. I. (2019). Chapter 2: High Mountain Areas. IPCC Special Report on the Ocean and Cryosphere in a Changing Climate. IPCC Special Report on the Ocean and Cryosphere in a Changing Climate, 131-202. Huning, L. S., & AghaKouchak, A. (2020). Global snow drought hot spots and characteristics. Proceedings of the National Academy https://doi.org/10.1073/pnas.1915921117Jasechko, of Sciences, 201915921. S. (2019). Global Isotope Hydrogeology - Review. Review of Geophysics, 57, 835–965. https://doi.org/https://doi.org/10.1029/2018RG000627Jensen, M. E., Rob, D. C. N., & Franzoy, C. E. (1969). Scheduling irrigations National Conference on Water Resources using climate-crop-soil data. Engineering of the American Society of Civil Engineers, 20 p.Julander, R. P., & Clayton, J. A. (2018). Determining the proportion of streamflow that is generated by cold season processes versus summer rainfall in Utah, USA. Journal of Hydrology: Regional Studies, 17(December 2017), https://doi.org/10.1016/j.ejrh.2018.04.005Koeniger, P., Hubbart, 36 - 46.J. A., Link, T., & Marshall, J. D. (2008). Isotopic variation of snow cover and streamflow in response to changes in canopy strucutre in a snow-dominated mountain catchment. Hydrological Processes, 22, 557–566. https://doi.org/doi: 10.1002/hyp.6967LANDFIRE. (2015). Existing vegetation type and cover layers. U.S. Department of the Interior, Geological Survey. http://landfire.cr.usgs.gov/viewer/(accessed May 2017)Leavesley, G. H., Markstrom, S. L., Brewer, M. S., & Viger, R. J. (1983). The Modular Modeling System (MMS)—The physical process modeling component of a database-centered decision support system for water and power management. Water, Air, and Soil Pollution, 90, 303-3011.Lee, J., Feng, X., Faiia, A. M., Posmentier, E. S., Kirchner, J. W., Osterhuber, R., & Taylor, S. (2010). Isotopic evolution of a seasonal snowcover and its melt by isotopic exchange between liquid water and ice. Chemical Geology, 270(1-4), 126-134. https://doi.org/10.1016/j.chemgeo.2009.11.011Marchetti, D. W., & Marchetti, S. B. (2019). Stable isotope compositions of precipitation from Gunnison, Colorado 2007–2016: implications for the climatology of a high-elevation valley. Heliyon, 5(7), e02120. https://doi.org/10.1016/j.heliyon.2019.e02120Markstrom, S. L., Regan, R. S., Hay, L. E., Viger, R. J., Webb, R. M. T., Payn, R. A., & LaFontaine, J. H. (2015). PRMS-IV, the precipitation-runoff modeling system, version 4. U.S. Geological Survey Technical Methods, Book 6 (Chapter B7), 157. https://doi.org/http://dx.doi.org/10.3133/tm6B7Milly, P. C. D., Betancourt, J., Falkenmark, M., Hirsch, R. M., Kundzewicz, Z. W., Lettenmaier, D. P., & Stouffer, R. J. (2008). Climate change. Stationarity is dead: whither water management? Science (New York, N.Y.), 319(5863), 573–574. https://doi.org/10.1126/science.1151915Molotch, N. P., Brooks, P. D., Burns, S. P., Litvak, M., Monson, R. K., McConnell, J. R., & Musselman, K. (2009). Ecohydrological controls on snowmelt partitioning in mixed-conifer sub-alpine forests. Ecohydrology, 2, 129–142. https://doi.org/DOI: 10.1002/eco.48Mott, R., Vionnet, V., & Grünewald, T. (2018). The Seasonal Snow Cover Dynamics: Review on Wind-Driven Coupling Processes. Frontiers in Earth Science, https://doi.org/10.3389/feart.2018.00197Musselman, K. N., 6(December). Molotch, N. P., & Brooks, P. D. (2008). Effects of vegetation on snow accumulation and ablation in a mid-latitude sub-alpine forest. Hydrological Processes, 22(15), 2767–2776. https://doi.org/10.1002/hyp.7050Oiro, S., Comte, J. C., Soulsby, C., & Walraevens, K. (2018). Using stable water isotopes to identify spatio-temporal controls on groundwater recharge in two contrasting East African aquifer systems. Hydrological Sciences Journal, 63(6), 862–877. https://doi.org/10.1080/02626667.2018.1459625OSU. (2012). PRISM Climate Group. http://prism.oregonstate.eduOtte, I., Detsch, F., Gütlein, A., Scholl, M., Kiese, R., Appelhans, T., & Nauss, T. (2017). Seasonality of stable isotope composition of atmospheric water input at the southern slopes of Mt. Kilimanjaro, Tanzania. Hydrological Processes, 31(22), 3932–3947. https://doi.org/10.1002/hyp.11311Painter, T. H., Berisford, D. F., Boardman, J. W., Bormann, K. J., Deems, J. S., Gehrke, F., Hedrick, A., Joyce, M., Laidlaw, R., Marks, D., Mattmann, C., McGurk, B., Ramirez, P., Richardson, M., Skiles, S. M. K., Seidel, F. C., & Winstral, A. (2016). The Airborne Snow Observatory: Fusion of scanning lidar, imaging spectrometer, and physically-based modeling for mapping snow water equivalent and snow albedo. Remote Sensing of Environment, 184, 139–152. https://doi.org/10.1016/j.rse.2016.06.018Sexstone, G. A., Clow, D. W., Fassnacht, S. R., Liston, G. E., Hiemstra, C. A., Knowles, J. F., & Penn, C. A. (2018). Snow Sublimation in Mountain Environments and Its Sensitivity

to Forest Disturbance and Climate Warming. Water Resources Research, 54(2), 1191-1211. https://doi.org/10.1002/2017WR021172Sinclair, K. E., & Marshall, S. J. (2008). Post-depositional modification of stable water isotops in winter snowpacks in the Canadian Rocky Mountains. Annals of Glaciology, 49, 96-106. https://doi.org/10.3189/172756408787814979Sprenger, M., Leistert, H., Gimbel, K., & Weiler, M. (2016). Illuminating hydrological processes at the soil-vegetation-atmosphere interface with water stable isotopes. Reveiw of Geophysics, 54, 674–704. https://doi.org/doi: 10.1002/2015RG000515Stichler, W., Rauert, W., & Martinec, J. (1981). Environmental isotope studies of an alpine snowpack (Weissfluhsoch Switzerland). Nordic Hydrology, 12(4-5), 297-308. https://doi.org/10.2166/nh.1981.0024Tappa, D. J., Kohn, M. J., McNamara, J. P., Benner, S. G., & Flores, A. N. (2016). Isotopic composition of precipitation in a topographically steep, seasonally snow-dominated watershed and implications of variations from the Global Meteoric Water Line. Hydrological Processess, 30, 4582–4592. https://doi.org/doi:10.1002/hyp.10940Taylor, S., Feng, X., Williams, M., & McNamara, J. (2002). How isotopic fractionation of snowmelt affects hydrograph separation. Hydrological Processes, 16(18),3683 - 3690.https://doi.org/https://doi.org/10.1002/hyp.1232Taylor, Susan, Feng, X., Kirchner, J. W., Osterhuber, R., Klaue, B., & Renshaw, C. E. (2001). Isotopic evolution of a seasonal snowpack and its melt. Water Resources Research, 37(3), 759-769. https://doi.org/10.1029/2000WR900341Taylor, Susan, Feng, X., Renshaw, C. E., & Kirchner, J. W. (2002). Isotopic evolution of snowmelt 2. Verification and parameterization of a one-dimensional model using laboratory experiments. Water Resources Research, 38(10), 36-1-36-38. https://doi.org/10.1029/2001wr000815Unnikrishna, P. V., McDonnell, J. J., & Kendall, C. (2002). Isotope variations in a Sierra Nevada snowpack and their relation to meltwater. Journal of Hydrology, 260(1-4), 38-57. https://doi.org/10.1016/S0022-1694(01)00596-0Zhang, Q., Knowles, J. F., Barnes, R. T., Cowier, R. M., Rock, N., & Williams, M. W. (2018). Surface and subsurface water contributions to streamflow from a mesoscale watershed in complex mountain terrain. Hydrological Processes. https://doi.org/doi:10.1002/hyp.11469

#### Figure Captions.

Figure 1. The East River (ER) watershed, Colorado (a) elevation with inset showing the ER in the context of the Colorado River Basin in the southwestern United States. (b) spatial distribution of ecozones based on elevation and dominant vegetation type. (c) Simulated snow water equivalent (SWE) near peak accumulation for an average water year (d) Spatial distribution of volumeweighted <sup>18</sup>O annual inputs from snowmelt and rain for an average water year (2016).

Figure 2. Subalpine (3361 m) daily (a) snow water equivlant (SWE), snow and rain, (b) snowmelt and <sup>18</sup>O influx, (c) <sup>18</sup>O influx as a function of snowmelt from initial melt to SWE=0. Montane daily (2620 m) (d) SWE, (e) snowmelt and <sup>18</sup>O influx, (f) <sup>18</sup>O influx as a function of snowmelt from initial melt to

SWE=0 Influx is defined as snowmelt or rain.

Figure 3. Percent change in d-excess snowmelt due to snowpack kinetic fractionation (a) dry/warm year, 2018, (b) wet/cool year, 2019, and (c) difference between 2019 and 2018, with negative percentages indicating a reduction in kinetic fractionation and positive precentages an increase. (d) The average annual ratio of potential evaportranspiration to precipitation (PET/P).